Author's response to reviews

Title: Determinants of rapid weight gain during infancy: baseline results from the NOURISH randomised controlled trial.

Authors:

Seema Mihrshahi (smihrshahi@yahoo.com)
Diana Battistutta (d.battistutta@qut.edu.au)
Anthea Magarey (Anthea.Magarey@flinders.edu.au)
Lynne A Daniels (l2.daniels@qut.edu.au)

Version: 4 Date: 4 November 2011

Author's response to reviews: see over
Dear Editor,

Re: MS ID : 1229257502524037

We thank you for the formatting changes that you recommended we have made the changes as outlined below. We have attached a copy of the final manuscript. We now believe that the paper is ready for publication:

**Major revisions (we require the author to make these changes)**

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
We have made these changes

Additional files: Please remove from the manuscript.
We have removed additional files

**Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)**

References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
We have reformatted the references

Tables: Please remove vertical lines.
We have removed the vertical lines

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We have checked the manuscript for typographical errors

Many thanks
Seema Mihrshahi